Author's response to reviews

Title: Epidemiologic studies of modifiable factors associated with cognition and dementia: Systematic review and meta-analysis

Authors:

May A Beydoun (baydounm@mail.nih.gov)
Hind A Beydoun (baydouha@evms.edu)
Alyssa A Gamaldo (gamaldoaa@mail.nih.gov)
Alisson Teel (TeelAL@evms.edu)
Alan B. Zonderman (zondermana@mail.nih.gov)
Youfa Wang (youfawan@buffalo.edu)

Version: 12 Date: 7 May 2014

Author's response to reviews: see over
April 11th, 2014

Dear Editor:

We are very pleased that you are considering our review paper “Epidemiologic studies of modifiable factors associated with cognition and dementia: Systematic review and meta-analysis” for possible publication in the journal BMC Public Health. We found the reviewers’ comments very helpful in improving the quality of the manuscript as a whole. Attached are the response to the reviewers’ comments and the revised text, tables, and figures with highlighted changes. We also added the PRISMA checklist in this revision.

We look forward to hearing from you soon regarding the final decision. Please do not hesitate to contact me for any questions,

Sincerely yours,

May A Beydoun
NIH Biomedical Research Center
National Institute on Aging, IRP
251 Bayview Blvd.
Suite 100
Room #: 04B118
Baltimore, MD 21224
E-mail: baydounm@mail.nih.gov
Fax: 410-558-8236